Author's response to reviews

Title: Shape (But Not Volume) Changes in the Thalami in Parkinson Disease

Authors:

Martin J McKeown (mmckeown@interchange.ubc.ca)
Ashish Uthama (ashishu@ece.ubc.ca)
Rafeef Abugharbieh (rafeef@ece.ubc.ca)
Samantha Palmer (sijpalmer@interchange.ubc.ca)
Mechelle Lewis (lewism@neurology.unc.edu)
Xuemei Huang (xuemei@med.unc.edu)

Version: 3 Date: 25 March 2008

Author's response to reviews: see over
March 23, 2008

Re: MS: 3135447016273307 “Shape (But Not Volume) Changes in the Thalami in Parkinson Disease”

Dear Editor,

As you advised, we have now included a "competing interests" section in the manuscript.

We trust that the manuscript is now suitable for publication.

Sincerely,

[Signature]

Martin J. McKeown
Ashish Uthama
Rafeef Abgharbieh
Samantha Palmer
Mechelle Lewis
Xuemei Huang